# Journal of Asian Scientific Research

ISSN(e): 2223-1331 ISSN(p): 2226-5724 DOI: 10.55493/5003.v15i3.5512

Vol. 15, No. 3, 316-339.

© 2025 AESS Publications. All Rights Reserved. URL: www.aessweb.com

# Examining the role of board characteristics in the CSR-dividend nexus: **Evidence from European firms**



Abdelaziz Hakimi¹+

Hichem Saidi²

Soufiene Tebassi<sup>3</sup>

1,3 V.P.N.C and Faculty of Law, Economics, and Management of Jendouba, University of Jendouba, Tunisia.

<sup>1</sup>Email: <u>abdelaziz.hakimi@fsjegj.rnu.tn</u>

<sup>3</sup>Email: <u>soufiene.tabessi@fsjegj.rnu.tn</u> <sup>2</sup>Department of Economics, College of Business, Imam Mohammad Ibn Saud Islamic University (IMSIU), Saudi Arabia.



<sup>2</sup>Email: <u>halsaidi@imamu.edu.sa</u>

#### Article History

Received: 20 May 2025 Revised: 14 July 2025 Accepted: 21 July 2025 Published: 4 August 2025

### **Keywords**

Board characteristics CSR practices Dividends ESG scores European firms SGMM.

#### **ABSTRACT**

Understanding how a company's board characteristics influence the connection between corporate social responsibility and dividend policies has become important. As more stakeholders become concerned about companies' ethical behavior, businesses are feeling increased pressure to balance their social responsibilities with financial rewards. This paper examines the relationship between corporate social responsibility and dividend payouts in European firms, with a focus on the moderating role of board characteristics. To achieve this goal, we examined a sample of 1,376 publicly listed European companies over the period from 2014 to 2023, using dynamic panel data regressions, specifically the System Generalized Method of Moments (SGMM). The findings suggest that having strong corporate social responsibility performance significantly influences the decision to distribute dividends. The study also shows that board characteristics moderate the relationship between corporate social responsibility performance and dividend payouts. These insights carry substantial policy implications. Studying the CSR-dividend relationship with board characteristics as a moderator highlights the board's critical role in aligning social responsibility with shareholder interests. It supports policies promoting board independence, diversity, and expertise to enhance governance quality. This can encourage responsible dividend strategies that reflect both financial performance and ethical commitments. Policymakers may use these insights to refine corporate governance regulations and CSR disclosure standards.

Contribution/ Originality: Studying the relationship between CSR and dividend payout in European firms offers some novelty by addressing a region with strong regulatory frameworks and stakeholder-oriented governance. It highlights how CSR performance influences financial policies in socially responsible markets. Additionally, examining the moderating role of board characteristics adds depth by linking corporate governance to strategic payout decisions.

#### 1. INTRODUCTION

Corporate social responsibility is recognized as an important topic across many fields, such as management, finance, and public relations. It's a key factor in helping companies build trust with their stakeholders as argued by Carroll [1], Freeman [2], Fourati and Dammak [3], and Quang and Le [4]. In addition, businesses are making CSR a central part of their strategies. Their goal is to address social issues and ensure they remain sustainable in the long term [5].

This change demonstrates how companies do not only pursue financial objectives. They also have a responsibility to help society and protect the environment [6]. As a result, corporate social responsibility has become a significant part of how companies operate today. It is no longer just about profits; it is also about how businesses impact the world around them. This shift has led to new standards in accounting and the development of environmental, social, and governance (ESG) scores, which are used to measure how well companies perform in these areas [7-9].

In recent years, there's been a shift in how companies are seen in society. People consider firms as responsible entities that should work to promote fairness and balance in both social and economic issues, all while looking out for their stakeholders, including shareholders, based on the idea of stakeholder theory, Jensen and Meckling [10] and Rodriguez-Fernandez [11]. CSR is becoming more than a buzzword; it is a way for companies to be more transparent and demonstrate their social responsibilities. Additionally, incorporating Environmental, Social, and Governance practices into their core strategies is becoming essential for ensuring that a company's growth is sustainable, ethical, and focused on the long term [12].

Besides CSR, good corporate governance makes companies more transparent, trustworthy, and reputable. It also provides important information that helps investors make smarter choices. When a company has a strong governance setup, it reduces risks and supports long-term success by ensuring that managers' interests align with those of shareholders and other interested parties [13]. The question of how CSR practices are integrated into a company's overall strategy often depends significantly on the specific composition of the board of directors. For example, having women on the board can make a difference because they might bring different viewpoints and pay more attention to social and environmental issues when making strategic decisions [14]. Besides, the financial knowledge of board members is important for successfully incorporating ESG factors and understanding how they affect the company's financial results [15, 16].

When the CEO also serves as the chairman of the board, it often leads to a more centralized approach to decision-making. This structure can influence how the company manages its social responsibility efforts, including aspects such as paying dividends [17]. Board diversity is crucial, as it enables companies to more effectively address global issues related to CSR and ESG. When a team includes individuals from diverse backgrounds, it becomes easier to implement these practices in a way that is comprehensive and effective across different cultures and perspectives [18, 19]. Furthermore, the size of the board also plays an important role. If it is too small, it may lack sufficient diversity of perspectives. Conversely, a board that is too large could slow down decision-making and hinder the smooth integration of ESG practices into the company's overall strategy [20].

Although companies that adopt CSR practices appear to enjoy greater social legitimacy and stakeholder engagement, the specific ways in which board characteristics moderate the impact of these practices on dividend payouts remain largely underexplored. In particular, it is important to understand how board composition, including the presence of women, financial expertise, CEO duality, cultural diversity, and board size, affects not only companies' CSR commitments but also their dividend distribution strategies. This raises the following question: How do board characteristics moderate the relationship between CSR and dividend distribution policy? The main objective of this paper is to analyze this relationship by exploring the mechanisms through which board composition moderates the CSR–dividend distribution relationship in European firms. To do so, we used a large sample of 1,376 listed companies located in 23 European countries over the period from 2014 to 2023 and applied the SGMM method as an econometric approach.

The European context presents a particularly relevant case study for this paper for several reasons. First, strict regulations in Europe impose transparency obligations regarding CSR, notably through the Corporate Sustainability Reporting Directive (CSRD). Second, societal and institutional pressures strongly encourage companies to integrate CSR into their strategies, in response to growing expectations from investors and regulators. Furthermore, Europe's structured and rigorous environment provides an adequate research framework for analyzing the relationship between governance and CSR commitment. In this context, the characteristics of boards of directors play a key role in enabling European companies to reconcile societal performance requirements

with a consistent dividend distribution policy. Indeed, factors such as board diversity, financial expertise, CEO duality, and board size can influence how companies balance the expectations of stakeholders and shareholders.

This study contributes to the existing literature in several ways. First, it addresses the research gap regarding the relationship between board characteristics, CSR, and dividend payout, with a focus on how governance influences these decisions. Recent studies have shown that board diversity, independence, and expertise separately impact CSR practices and dividend policies [21-23]. However, no prior studies have investigated the joint effect or the moderating role. Second, the results of this study provide valuable insights for corporate managers, investors, and regulators. These results lead to a better understanding of the impact of board characteristics on dividend distribution decisions in the context of CSR practices. Finally, this research explores the moderating role of corporate governance between CSR and dividend payout, offering implications for regulators and policymakers.

The paper is organized as follows: Section 2 reviews the theoretical literature. Section 3 presents the empirical literature review and formulates the hypotheses. Section 4 describes the empirical design. The empirical results and robustness checks are discussed in Section 5. Finally, Section 6 concludes and offers some policy recommendations.

### 2. THEORETICAL LITERATURE REVIEW

Theoretically, the relationship between CSR and dividend policy is supported by several theories. First, the agency theory explains the link between CSR and dividend policies by emphasizing conflicts that can arise between managers and shareholders. According to Jensen and Meckling [24], managers may engage in CSR as a means of self-promotion or to enhance their reputation, potentially at the expense of shareholder wealth. In this context, paying dividends can help resolve some issues that occur when managers and shareholders have differing interests. By distributing dividends, companies can prevent managers from using excessive free cash flow for their own purposes. For companies that prioritize social responsibility, paying dividends can also reassure investors that the company's resources are being used responsibly, which helps ensure alignment between managers and shareholders [25]. Hence, CSR and dividend policies could be viewed as two tools that work together to help reduce agency problems and improve how a company is run.

Second, the *signaling theory* suggests that when companies pay dividends or disclose CSR efforts, they transmit signals indicating that they are financially stable and have sound plans. In the presence of information asymmetry, where investors cannot fully observe a firm's internal operations, companies may use CSR activities and dividend payouts to convey positive information. High CSR engagement signals that a firm is socially responsible and committed to long-term sustainability, while consistent or increasing dividend payments indicate stable cash flows and profitability [26]. When used together, CSR and dividends enhance investor confidence and market valuation, as they collectively signal ethical management and financial robustness.

Third, the *stakeholder theory* broadens the focus beyond shareholders to consider the interests of all stakeholders, including employees, customers, communities, and the environment. According to Freeman [2], firms are expected to create value not just for shareholders but for all stakeholders. In this framework, CSR becomes a strategic tool to manage stakeholder relationships and maintain social legitimacy. However, distributing dividends to shareholders must be balanced with investments in stakeholder-related initiatives. This tension may influence dividend policy, as firms with strong CSR commitments might retain more earnings to fund socially responsible projects, potentially lowering dividend payouts [27]. Thus, stakeholder theory supports a more cautious and stakeholder-sensitive approach to dividend decisions.

Fourth, the *resource-based view* (RBV) of the firm emphasizes the role of unique internal resources and capabilities in creating a competitive advantage. CSR activities can be considered intangible assets that enhance a firm's reputation, trustworthiness, and stakeholder loyalty factors that contribute to long-term profitability [28]. Under this view, firms may strategically adjust dividend policies to invest more heavily in CSR, viewing it as a long-term investment rather than a cost. The implication is that lower dividends today may be justified by greater

value creation through enhanced brand equity and stakeholder goodwill in the future. Therefore, RBV supports a trade-off between immediate shareholder returns and long-term strategic investments in CSR.

### 3. EMPIRICAL LITERATURE REVIEW AND DEVELOPMENT OF HYPOTHESES

In a changing economic landscape, CSR has emerged as a key strategic factor, gaining increasing prominence in business decisions. According to Carroll [29], CSR refers to a company's commitment to ethical practices, while promoting economic development and improving the quality of life for employees, their families, and society as a whole. CSR is now seen as a way to create value over the long term [30]. At the same time, dividends continue to be a key factor for investors because they show how profitable a company is and can influence investment choices [31]. Paying dividends is often viewed as a sign of a company's financial strength and its growth potential [32].

Recent studies have looked into how CSR relates to a company's dividend policies. Several research efforts, like those by Dahlia [33], Sheikh et al. [34], and Dewasiri and Abeysekera [35] have found a positive link between CSR activities and how companies handle dividends. Companies that prioritize good CSR practices tend to pay out dividends more regularly, partly to boost their image with investors and to make financial information more transparent. However, Abdel-Wanis [36] points out that this relationship changes depending on where a company is in its life cycle.

For the US, Cheung et al. [16] discovered that companies with strong CSR practices tend to pay higher dividends. In Europe, Samet and Jarboui [37] noted that firms with better CSR performance are more likely to follow generous dividend policies. According to stakeholder theory, engaging in CSR can bring multiple benefits to a company, especially when it comes to increasing revenue and managing costs Ongsakul et al. [38] and Chouaibi et al. [39]. Sheikh et al. [34] concluded the same results in Pakistan. Additionally, Dewasiri and Abeysekera [35] observed similar results in Sri Lanka. Overall, these studies suggest that CSR tends to boost a company's dividend payouts worldwide, which might help reduce agency issues and make companies more transparent to their stakeholders. Nevertheless, some researchers point out that companies committed to CSR often choose to retain more profits to fund social or environmental projects, which can limit dividend payouts to shareholders [40].

#### 3.1. CSR and Dividend Policy: The Moderating Role of Board Characteristics

The board of directors is considered a central component of the corporate governance framework. Corporate governance refers to the set of mechanisms through which investors ensure a return on their investments [41-43]. Within this context, the board plays a crucial role in supervising and guiding corporate strategy [44-46]. Several studies have highlighted that the effectiveness of a board depends on various characteristics, with its size being one of the most critical. Indeed, board size is considered a key factor influencing decision-making efficiency and oversight capabilities [20, 47, 48].

#### 3.1.1. Board Size

Larger boards may better reflect the diversity of stakeholder interests, as suggested by Al-Saidi [49]. However, having a larger board can sometimes exacerbate agency problems. As discussed in the seminal work of Jensen and Meckling [24]. Larger boards may grant the CEO greater influence over decision-making and the company's direction. This can result in less effective oversight, which may compromise the overall quality of corporate governance [50].

Recent research by Awad et al. [51] suggests that larger boards can make it more difficult to maintain alignment on priorities and strategic direction, particularly in complex organizations. Despite these challenges, larger boards can still be highly effective. With more members, tasks can be distributed more evenly, and monitoring responsibilities, especially those related to CSR, tend to be carried out more efficiently. Several studies

also highlight that larger boards may enhance the impact of CSR on financial performance by enabling better management of CSR initiatives and providing greater resources for oversight and strategic advice [52, 53].

Prior researches conclude that the board size could impact both corporate social responsibility and dividend payments. For example, Kalsie and Shrivastav [54] found that a larger board might lead to greater efficiency, help reduce agency problems, and encourage companies to pay dividends. Similarly, Beji et al. [55] found a positive link between board size and CSR activities. [34] emphasized that board size can also play an important role in how dividends are distributed. Given these different perspectives, it seems reasonable to adopt a balanced view of how board size affects these areas. Based on this, we propose the following hypotheses:

Hypothesis 1: There is a positive relationship between board size and dividend payout.

Hypothesis 1a: The relationship between CSR and dividend payout could be moderated by board size.

### 3.1.2. CEO Duality

CEO duality occurs when one person holds both the CEO and chairperson roles. This board characteristic has been the subject of many recent research studies [56-58]. When these roles are combined, they give the executive significant power over the company's strategy and decisions, including those related to profits and investments [59]. Some studies suggest that this combination can enhance unity and speed up decision-making [60] while others warn it may lead to issues such as reduced oversight and excessive influence by the CEO [61]. Therefore, whether CEO duality benefits or harms a company largely depends on the overall governance structure and the specific context of the company [62].

The relationship between CEO duality and dividend payouts is quite complex and has led to mixed results. Some studies suggest that when the CEO also serves as the chair, it can create a clearer strategic direction and lead to better dividend decisions [63, 64]. However, others argue that concentrating on these roles might result in decisions that are not fully aligned with shareholders' interests. Understanding this relationship requires careful consideration of each specific context [64, 65].

In addition, the role of the CEO is evolving as companies focus on social responsibility. Today, creating both financial and social value is a core part of corporate strategy. Research on the interaction between CEO duality and CSR has produced mixed results. Some studies suggest that when the same person holds both the CEO and chair roles, it can reduce transparency and corporate engagement in social issues [17, 56, 66, 67]. However, more recent research indicates that CEO duality can enhance CSR disclosure and initiatives, particularly when stable leadership aligned with strategy is essential [68, 69]. In emerging markets, granting strong authority to the CEO can help advance long-term CSR efforts, especially when supported by sound internal governance [70, 71]. These findings suggest that the effects of CEO duality are complex and context-dependent, highlighting the need for further research. Based on this, we propose the following hypotheses:

Hypothesis 2: There is a positive relationship between CEO duality and dividends paid.

Hypothesis 2a: The relationship between CSR and dividend payments may be moderated by CEO duality.

# 3.1.3. Board Member Independence

The presence of independent directors on the board is generally regarded as an important indicator of good corporate governance [72, 73]. It encourages management to act in the best interests of shareholders, fairly represent other stakeholders, and reduce information asymmetry within the organization [74]. Board independence is also considered essential under the *agency theory*. This theory argues that independent directors should hold the majority of board seats to strengthen oversight over management, as emphasized by Fama and Jensen [75]. This independence enhances board effectiveness and reduces agency costs by compelling management to distribute a greater portion of excess cash in the form of dividends [76].

The results of prior research exploring how independent members influence firm decisions and dividend policies are mixed. Many recent studies, such as those by Riaz et al. [77], Alshabibi et al. [78], and Kilincarslan [79], suggest that having a more independent board can lead to better governance and higher dividend payouts. Similarly, Shahid et al. [80] and Al Farooque et al. [81] found that increased board independence tends to be associated with more frequent or higher dividends. Interestingly, research by Al Shabibi and Ramesh [82] and Ranti [83] points out that this effect may be even more pronounced in family-owned companies, where independent directors help strengthen governance and support more generous dividend strategies.

From another perspective, the literature also highlights the importance of board independence in promoting CSR. The studies of Islam et al. [84] and Agarwala et al. [85] have shown that there is often a positive and meaningful connection between an independent board and CSR efforts. Independent directors tend to be less influenced by internal pressures and are generally more focused on transparency and ethical practices, making them key players in guiding companies toward socially responsible behavior. Supporting this view, Al Fadli et al. [86] emphasize that independent boards are critical in improving CSR reporting. Given these perspectives, we propose the following hypotheses regarding the role of board independence:

Hypothesis 3: An independent board positively affects dividend payout.

Hypothesis 3a: The relationship between CSR and dividend payout could be moderated by the level of board independence.

### 3.1.4. Gender Diversity

The increasing presence of women on corporate boards demonstrates how companies are progressing toward more inclusive leadership. Having diverse boards helps companies make better, more well-rounded decisions [18]. In this case, women's representation is especially important in shaping a company's strategies, such as its approach to corporate social responsibility (CSR) and dividend policies [87].

Recent research supports a strong relationship between the size of dividend payouts and the diversity of company boards, including female directors [88, 89]. The increasing presence of women in leadership positions is associated with improved governance quality and the promotion of more prudent and stable dividend policies.

As gender diversity continues to grow, especially with more women taking seats on corporate boards, we are witnessing a genuine shift towards more comprehensive and balanced governance. Having a diverse mix of board members can lead to better decision-making because different perspectives help create more well-rounded strategies [18]. In this context, women on boards often influence company direction, especially regarding CSR and dividend policies [87]. Recent research suggests a strong link between gender diversity on boards and how companies handle dividend payouts [89]. The more women leaders there are, the better the quality of governance tends to be, which often results in more careful and stable dividend decisions. Female directors are generally seen as more riskaverse [90], meaning they tend to support responsible and steady dividends [91]. This cautious stance helps reduce conflicts between managers and shareholders and limits managers from acting opportunistically. Besides, having women on boards promotes stronger CSR efforts. A high female presence is often viewed as a positive sign of a company's ESG performance [92]. This shows that gender diversity influences strategic decisions well beyond just financial matters. However, the link between board gender diversity and dividend policy isn't straightforward. Some researchers, like Gharbi and Jarboui [937], have found that more women on boards might decrease dividend payouts. Others, like García-Meca et al. [94], suggest the relationship isn't linear. It might follow an inverted Ushape, where initial increases in female representation boost dividends, but beyond a certain point, they start to decline again. This indicates that the relationship is complex and depends on many factors within each company. While having more women on boards tends to encourage careful financial strategies and better CSR, its effect on dividend payments can vary depending on the specific context. Based on these findings, we propose the following hypotheses:

Hypothesis 4: There is a positive relationship between gender diversity and dividend payout.

Hypothesis 4a. The relationship between CSR and dividend payout could be moderated by gender diversity.

### 3.1.5. Cultural Diversity

Having cultural diversity on company boards adds value to how companies are run. It helps bring different perspectives to decision-making and makes it easier to handle social, economic, and environmental challenges effectively [95, 96]. Diversity on boards, including differences in nationality, professional backgrounds, gender, and perspectives, helps create more balanced and comprehensive decision-making. When boards are diverse, they are often more engaged in corporate social responsibility (CSR) and make financial decisions, such as dividend distribution, with a focus on long-term sustainability. Recent research confirms that such diversity not only improves how a company is governed and its CSR efforts but also positively impacts how it handles dividends [18, 97-99]. Based on our earlier discussion, having a diverse group of people on company boards significantly contributes to good governance. Recent research indicates that diversity enhances transparency, promotes more comprehensive decision-making, and increases companies' engagement in social and environmental initiatives, particularly regarding sustainability [100, 101]. It also affects how companies handle dividend payments. When board members come from different backgrounds, they tend to be more careful and balanced in their decisions, focusing on what is best for the company and everyone involved in the long term [102]. Other studies also found that diversity helps companies better prepare for global challenges and creates stronger, more flexible governance structures [18]. Based on these findings about cultural diversity, we have formulated the following hypotheses:

Hypothesis 5: There is a positive relationship between cultural diversity and dividend payout.

Hypothesis 5a. The relationship between CSR and dividend payout could be moderated by cultural diversity.

### 3.1.6. Financial Expertise

Board members with strong financial knowledge are important when balancing CSR efforts and dividend policies. When board members understand finance well, they can better assess how CSR activities will impact the company's finances, helping the company weigh short-term profits against long-term social benefits. Many studies support this idea. For example, Naheed et al. [103] found that in China, companies with financially knowledgeable board members tend to be more transparent about their CSR efforts. Similarly, Lin and Nguyen [56] and Zhe et al. [104] demonstrated that CFOs with more experience are more likely to be involved in CSR activities than those with less experience. When it comes to dividend policy, studies show that the effects can depend on the context. For instance, Thompson and Manu [105] found that in China, having board members with strong financial expertise tends to support the idea that dividends can be a way to reduce agency problems. But in Pakistan, their results suggest the opposite, that dividends help complement efforts to align management and shareholder interests. Similarly, research by Sarwar et al. [106] and Wu et al. [107] emphasizes how in Kenya, the level of financial expertise on a company's board plays a key role in shaping dividend policies. Also, El Ghoul et al. [108] discovered that boards with more financial know-how are better at balancing the trade-offs between pursuing social responsibility initiatives and paying dividends, leading to clearer and more balanced corporate governance. Based on this evidence emphasizing the importance of financial expertise, we propose these hypotheses:

 $\label{thm:continuous} \textit{Hypothesis 6: Financial expertise on the board increases dividend payouts.}$ 

Hypothesis 6a: The relationship between CSR and dividend payout could be moderated by the level of financial expertise on the board.

### 4. EMPIRICAL DESIGN

#### 4.1. The Sample

To explore how board characteristics moderate the relationship between CSR and dividend payout, we used an initial sample of 1,405 companies across 23 European countries. However, 29 companies were excluded due to

missing data on board members' gender diversity, cultural backgrounds, or specific skills. The study covered the period from 2014 to 2023. All the information used in this research was sourced from Refinitiv's Eikon database, which provides detailed data on more than 72,000 publicly listed companies worldwide. This database includes various aspects such as dividend distribution, gender and cultural diversity on boards, financial expertise of board members, company size, CEO duality, ESG scores, and company fundamentals. Table 1 presents the number of firms per country.

Table 1. Number of companies per country.

Countries	es Number of firms % Countries		Number of firms	%	
Austria	46	3.34	Italy	45	3.27
Belgium	76	5.52	Luxembourg	34	2.47
Cyprus	39	2.83	Malta	30	2.18
Czech Republic	50	3.63	Netherlands	61	4.43
Denmark	63	4.58	Norway	74	5.38
Finland	62	4.51	Poland	66	4.80
France	61	4.43	Portugal	60	4.36
Germany	74	5.38	Spain	77	5.60
Greece	30	2.18	Sweden	96	6.98
Hungary	40	2.91	Switzerland	87	6.32
Iceland	36	2.62	United Kingdom	98	7.12
Ireland	71	5.16			
	Total	companies (v	vhole sample): 1.376		

### 4.2. Variable Selection

### 4.2.1. Dependent Variable

The dependent variable in this study is the dividend per share (DPS), which is the amount a company pays to its shareholders for each share they own. As Gitman and Zutter [109] explain, the DPS indicates how much profit is distributed to each shareholder for one common share.

### 4.2.2. Independent Variables

Among the independent variables, corporate social responsibility was measured using the ESG score. This proxy ranges from 0 to 1 and aggregates a company's performance across the three pillars: environmental, social, and governance. This approach makes it easier to compare companies by providing a clear overview of their sustainable and responsible impact [110]. The size of the board of directors (BS), defined as the total number of members, is an important factor in determining how effective the board is at overseeing strategic decisions [111]. The specific skills of board members (BSS), often measured by the percentage of directors with financial expertise, are associated with better decisions related to corporate responsibility and sustainability [112]. CEO duality (DUAL), which indicates whether the same person serves as both CEO and board chair, and board independence (IND), measured by the percentage of independent directors, are key to ensuring the separation of powers and minimizing conflicts of interest [113, 114]. Finally, cultural diversity (BCD), represented by the percentage of directors from different cultural backgrounds, and gender diversity (BGD), reflected by the percentage of women on the board, are considered important factors in enhancing governance and transparency [115].

As part of our robustness check, we developed an index of board characteristics (BC-index), which captures six important aspects of a company's board. These include board size (BS), board independence (IND), CEO duality (DUAL), the financial expertise of board members (BSS), gender diversity (BGD), and cultural diversity (CD). The structure and independence of the board can influence the effectiveness of company oversight and the level of transparency [116]. Meanwhile, diversity and expertise tend to enhance the quality of strategic decision-making [68]. When the CEO also serves as the board chair (DUAL), it can alter the balance of power within the company

[117]. Cultural diversity, in turn, brings a broader, more global perspective and can contribute to better risk management [118]. Table 2 presents the construction of the board characteristics index in detail.

Dimension	Indicator	Description	Adjustment	Weight
Size	Board size	Total number of board members	Standardized value	0.15
Diversity	Gender diversity	% women on the board of directors	Standardized value	0.20
Diversity	Cultural diversity	% of members from cultural minorities or foreign countries	Standardized value	0.15
Expertise	Financial expertise	% of members with financial training	Standardized value	0.20
Function	CEO duality	1 if the CEO is also Chairman of the Board, 0 otherwise.	Standardized value	0.1
Independence	Proportion of independent members	% of independent board members	Standardized value	0.20

Table 2. The main indicators used to construct the BC index.

After selecting the key indicators to analyze the board's features, the next step was to calculate their normalized values. To do this, we first identified the minimum and maximum values for each indicator. This allowed us to make fair comparisons between the indicators by placing them on a common scale. There are two main approaches to data normalization: statistical normalization and empirical normalization [119]. Statistical normalization adjusts each indicator so that they have a mean of zero and a standard deviation of one. This approach facilitates comparison by eliminating distortions caused by differences in the indicators' average values. This method is based on the following formula:

$$BC_{it} = \frac{BC \ i.t - \mu \ (BCi.t)}{\sigma \ (BC \ i.t)} \tag{1}$$

The value of BCi,t shows how the board characteristics compare after being scaled, with  $\mu(BCi,t)$  representing the average and  $\sigma(BCi,t)$  indicating the standard deviation. This approach ensures that the data remains within a common, understandable range. Conversely, empirical normalization assumes that a score of 1 indicates a more effective board, while a score near 0 indicates an ineffective board.

$$Normalized\ value = \frac{BC - Min\ (BC)}{Max\ (BC) - Min\ (BC)} \tag{2}$$

BC shows the current value of the characteristics listed in the table. The Min (BC) and Max (BC) represent the lowest and highest values observed for this indicator, respectively. This method of normalization makes it easier to compare different indices, even if they are measured in different units or on different scales. The composite index is then calculated using the following formula:

Board index = 
$$\frac{1}{n}\sum_{i=1}^{n} Normalized Values$$
 (3)

### 4.2.3. Moderating Variables

Let's examine how CSR interacts with different aspects of a company's governance, starting with the size of the board of directors (ESG\*BS). In our study, we focus on the total number of directors, which is a commonly used measure of board size in the literature, as seen in studies by Beji et al. [55], Jilani and Chouaibi [50], Sheikh et al. [34], and Chang et al. [120].

ESG\*IND is the interaction variable between CSR and the independence of the board of directors. We assess board independence by examining the proportion of independent directors relative to the total number of board members. This is a widely used measure in numerous studies, including those by Riaz et al. [77], Alshabibi et al. [78], Kilincarslan [79], and Narang et al. [76].

ESG\*DUAL is the interaction variable between CSR and CEO duality. We also examine CEO duality by determining whether the CEO also serves as the chair of the board, using a simple binary indicator: 1 if the roles are

combined and 0 if they are separate. This approach has been used in studies by Tang [60], Mubeen et al. [62], and Debnath et al. [61].

ESG\*BGD is the interaction variable between CSR and gender diversity on the board of directors. We also explore gender diversity by examining the percentage of women on the board, which reflects the level of gender balance within company governance. This approach is based on the work of Ye et al. [91], Bidi [121], and Wong et al. [92].

ESG\*BCD is the interaction variable between CSR and cultural diversity on the board of directors. Beyond gender, cultural diversity is also a key factor. We measure it by the percentage of board members whose cultural background differs from that of the company's headquarters. This approach is supported by studies from Adams and Ferreira [97], Laksmana et al. [98], Post and Byron [18], Khan and Yilmaz [99], and Aliani et al. [96].

ESG\*BSS is the interaction variable between CSR and financial expertise on the board of directors. We consider the financial expertise of board members, measured by the percentage of those with significant financial experience, to understand how financial skills contribute to corporate social responsibility efforts. This indicates the board's ability to make informed financial decisions. A high percentage suggests better competence in the company's financial management and has been applied in various studies: Sarwar et al. [106] and Wu et al. [107]; Naheed et al. [103]; Lin and Nguyen [56] and Zhe et al. [104].

### 4.2.4. Control Variables

Two control variables are included to account for the influence of company size and profitability on DIVPS. These are company size (SIZE), measured by the natural logarithm of its total assets, and return on assets (ROA), represented by the logarithm of the difference between the market value of assets and their book value. Analyzing these factors allows us to examine how board characteristics moderate the relationship between ESG and DIVPS. Table 3 shows the definition and measurement of the set of variables.

Table 3. Description of variables and measurement.

Variable	Description	Measurement
Dependent va	ariable	
DIPS	Dividend per share	Dividend per share (DPS) is the amount a company pays its shareholders for each share they own.
Independent	variables	
ESG score	Environmental, social, and governance	The ESG score is a global score based on self-reported information in the environmental, social, and corporate governance pillars.
Moderating v	U	
BCD	Board culture diversity	Percentage of board members from different cultural backgrounds
BGD	Board gender diversity	Percentage of women on the board of directors
BS	Size of the board of directors	Total number of board members
DUAL	CEO duality	The dummy variable that takes on the value 1 if the chairman of the board of directors of the company is simultaneously the CEO, or 0 otherwise.
IND	Board independence	The ratio of independent board members to the total number of board members.
BSS	Board-specific skills	The percentage of board members with financial experience
BC_index	Index of board characteristics	See the sub-section 3.2.2
Control varia	bles	
SIZE	Size of company	Natural logarithm of total assets
ROA	Return on assets	Natural logarithm of the ratio between the market value and the balance sheet value of assets.

**Note:** This table presents the definitions of the variables used in the context of our research.

### 4.3. Empirical Approach

In this study, we used the System GMM model, which is efficient for addressing endogeneity issues that often arise in company finance research. Endogeneity can cause biases in our estimates, leading to unreliable results. The SGMM method was first introduced by Blundell and Bond [122]. It helps to mitigate these problems by using internal instruments within the data itself. This makes the results more reliable compared to other methods like ordinary least squares (OLS) or fixed and random effects models, which can be affected by missing variables or measurement errors. Biases from missing variables and measurement mistakes are common in other estimation approaches, highlighting the importance of SGMM. It is widely used in finance and banking studies, especially when analyzing factors that influence bank performance [123] and how companies decide on their capital structures [124]. By effectively reducing the distortions caused by endogeneity, SGMM helps produce more accurate and consistent results, making it a go-to choice for analyzing panel data in corporate finance.

In this study, the empirical strategy investigates the moderating effect of board characteristics on the relationship between corporate social responsibility and dividend payout. The econometric model is provided in the following equation:

$$DIVPS_{i,t} = \beta_0 + \beta_1 DIVPS_{i,t-1} + \beta_2 ESG_{i,t} + \beta_3 \sum_{i=1}^{6} BC_{i,t} + \beta_4 \sum_{i=1}^{6} (ESG*BC)_{i,t} + \beta_5 ROA_{i,t} + \beta_6 SIZE_{i,t} + \epsilon_{i,t}$$
(4)

Where;

 $\sum_{i=1}^{6} BC$ : Is a matrix of board characteristics including BS, DUAL, IND, BGD, BCD, and BSS.

 $\sum_{i=1}^{6} ESG * BC$ : Is a matrix of interaction between CSR measured by ESG score and board characteristics.

# 5. EMPIRICAL RESULTS

# 5.1. Descriptive Statistics and Correlation Matrix

The descriptive statistics are provided in Table 4, highlighting significant variability in the characteristics of the analyzed companies. The Return on Assets (ROA) has an average of 4.58%, with considerable dispersion, indicating notable differences in company performance. The average company size is 21.44, with a standard deviation of 1.99, ranging from 8 to 27.13. Regarding ESG scores, there is high variability, with an average of 58.53 and a standard deviation of 18.30, suggesting that some companies are highly committed to these practices, while others are less so. The number of board members varies significantly, with an average of 9.98 and a standard deviation of 3.71, ranging from 3 to 30 members, which can influence decision-making processes and governance. CEO duality, though rare, has an average of 0.139 and shows significant variability, potentially impacting the company's governance structure. The proportion of independent members on boards of directors varies widely, with an average of 60.51% and a standard deviation of 21.75, reflecting boards with varying levels of independence. Regarding gender and cultural diversity, the averages are 30.95% and 31.64%, respectively, but the large standard deviations indicate a wide disparity in the representation of women and different cultures, with values ranging from 4.16% to 100%. Finally, financial expertise also shows considerable variability, with an average of 39.16% and a standard deviation of 19.97, indicating diverse levels of expertise among board members, which can influence financial decision-making.

Table 4. Summary statistics.

Variable	Obs.	Mean	Std. Dev.	Min.	Max.
ROA	13,760	4.587	14.0	-346.069	236.782
TAILLE	13,760	21.446	1.99	8.006	27.132
ESG	13,760	58.537	18.30	0.627	95.766
BS	13,760	9.984	3.71	1	30
DUAL	13,760	0.139	0.34	0	1
IND	13,760	60.513	21.75	3.33	100
BGD	13,760	30.95	11.85	4.16	100
BCD	13,760	31.64	24.15	4.34	100
BSS	13,760	39.167	19.97	3.57	100
ESG*BS	13,760	612.05	342.94	4.84	1965.83
ESG*DUAL	13,760	60.12	17.63	1.80	94.11
ESG*IND	13,760	60.51	21.75	3.33	100
ESG*BGD	13,760	1882.37	965.06	6.26	5717.09
ESG*BCD	13,760	2045.58	1719.29	62.67	9518.49
ESG*BSS	13,760	2272.94	1299.87	13.71	7695.28

Note: \* an interactional relationship between CSR and board characteristics.

Table 5 presents the correlation matrix. Pearson correlation coefficients were calculated to examine the associations among the independent variables. The Pearson correlation matrix does not show any correlation values equal to or greater than 0.8 [125]. The results indicate that there are no issues of multicollinearity in this analysis, as the correlations between the variables remain below 0.80. Moreover, all variables related to board characteristics exhibit a significant positive correlation.

## 5.2. Discussion of the Empirical Findings

In this subsection, we present and discuss the empirical results for the entire sample. Specifically, we interpret the results of the empirical strategy, which examines the moderating role of board characteristics on the relationship between the overall ESG score and dividend payout. We estimate six models, as we have six board characteristics. The empirical results obtained using the SGMM techniques are presented in Table 6.

 Table 5. Correlation matrix.

Variables	ROA	SIZE	ESG	BS	DUAL	IND	BGD	BCD	BSS	ESG*BS	ESG*DUAL	ESG*IND	ESG*BGD	ESG*BCD	ESG*BSS
ROA	1.0000														
SIZE	0.0611*														
	0.0000	1.0000													
ESG	-0.0366*	0.5857*													_
	0.0006	0.0000	1.0000												
BS	-0.0782*	0.5705*	0.4042*												
	0.0000	0.0000	0.0000	1.0000											
DUAL	0.0190*	0.2070*	0.0460*	0.1419*											
	0.0311	0.0000	0.0000	0.0000	1.0000										
IND	0.0043	0.0174	0.1923*	-0.2922*	-0.1615*										
	0.6960	0.1116	0.0000	0.0000	0.0000	1.0000									
BGD	0.0047	0.0235*	0.1779*	-0.0263*	0.0516*	0.0957*									
	0.6668	0.0318	0.0000	0.0160	0.0000	0.0000	1.0000								
BCD	0.0197	0.0133	0.0478*	-0.2539*	-0.0137	0.2260*	-0.0886*								
	0.1672	0.3509	0.0008	0.0000	0.3393	0.0000	0.0000	1.0000							
BSS	0.0615*	-0.1161*	-0.1070*	-0.2458*	-0.0158	-0.0083	-0.1276*	0.1294*							
	0.0000	0.0000	0.0000	0.0000	0.1486	0.4584	0.0000	0.0000	1.0000						
ESG*BS	-0.0718*	0.6884*	0.7685*	0.8680*	0.1164*	-0.0874*	0.0788*	-0.1566*	-0.2245*						
	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	1.0000					
ESG*DUAL	0.0032	0.5903*	1.0000*	0.3956*	0.2070*	0.1776*	0.1844*	0.0255	<b>-</b> 0.1411*	0.7661*					
	0.7864	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.3963	0.0000	0.0000	1.0000				
ESG*IND	0.0043	0.0174	0.1923*	-0.2922*	<b>-</b> 0.1615*	0.0633*	0.0957*	0.2260*	-0.0083	-0.0874*	0.1776*				
	0.6960	0.1116	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.4584	0.0000	0.0000	1.0000			
ESG*BGD	-0.0117	0.3581*	0.6842*	0.1987*	0.0634*	0.1692*	0.8063*	-0.0290*	-0.1425*	0.4800*	0.6605*	0.1692*			
	0.2871	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0475	0.0000	0.0000	0.0000	0.0000	1.0000		
ESG*BCD	0.0326*	0.2004*	0.3841*	-0.1330*	0.0006	0.2832*	-0.0237	0.9031*	0.0734*	0.0798*	0.3416*	0.2832*	0.1706*		
	0.0226	0.0000	0.0000	0.0000	0.7641	0.0000	0.1051	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	1.0000	
ESG*BSS	0.0248*	0.2035*	0.4426*	-0.0204	0.0025	0.1062*	-0.0142	0.1539*	0.7998*	0.1879*	0.4631*	0.1062*	0.2242*	0.2789*	,
	0.0229	0.0000	0.0000	0.0619	0.7161	0.0000	0.2070	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	1.0000

Note: \*, indicates the significance level at 5%.

Table 6. Regression results.

DIVPS	M (1)	M (2)	M (3)	M (4)	M (5)	M (6)
	-0.306	-0.311	-0.308	-0.309	-0.311	-0.308
DIVPS (-1)	-355.85***	-643.03***	-745.09***	-587.84***	-328.30***	-751.45***
	0.266	0.341	0.313	0.386	0.173	0.344
SIZE	4.290***	5.720***	5.590***	7.110***	2.850***	6.340***
	0.363	0.425	0.442	0.448	0.357	0.443
ROA	16.830***	21.790***	27.810***	26.060***	16.140***	27.360***
	0.075	0.070	0.040	0.039	0.014	0.013
ESG	5.360***	16.210***	8.330***	6.490***	3.060***	3.210***
	-0.020					
BS	-0.390					
na cuina	-0.003					
ESG*BS	-2.440**					
DILLI		-0.694				
DUAL		-0.590				
ECC*DIAI		-0.062				
ESG*DUAL		-3.420***	0.000			
IND			0.003			
IND			0.580			
ESG*IND			0.062 3.420***			
ESG-IND			3.420	0.056		
BGD				3.120***		
ESG*BGD				0.001 2.130**		
					0.155	
BCD					2.920	
					0.002	
ESG*BCD					0.430***	
						0.001
BSS						0.100
						0.001
ESG*BSS						2.530**
	-6.977	-8.789	-8.415	-9.561	<b>-</b> 4.949	-8.859
CONS	-5.510	-7.120	-7.400	-8.390	-3.940	-7.940
AR (1)	-1.0285	-1.0378	-1.0377	-1.0387	-1.0336	-1.0386
Prob	0.3037	0.2994	0.2994	0.2989	0.3013	0.2990
AR(2)	0.9962	0.9971	0.9964	0.9961	0.9955	0.9968
Prob	0.3191	0.3187	0.3190	0.3192	0.3195	0.3188
Sargan test	108.771	150.165	140.706	159.077	110.397	141.769
Prob	0.5122	0.4145	0.5927	0.5968	0.5958	0.5879
Obs	10321	10321	10321	10321	10321	10321

**Note:** \*\*\* and \*\*, indicate the significance level at 1% and 5%.

For model (1) regarding the impact of the interaction between BS and ESG on dividend payouts, we note that the variable DIVPS (-1) has a negative coefficient of -0.306. Each unit increase in DIVPS (-1) leads to a decrease of 0.306 in the present value of DIVPS. This result implies that the current dividend distribution is negatively associated with the dividend of the previous year.

In Model (1), results show that the coefficient of the interaction between ESG and board size (ESG\*BS) is negative and significant. This suggests that although ESG practices generally have a positive effect on dividend payout, this impact tends to diminish as board size increases. This trend could be explained by higher costs associated with management complexity or slower decision-making within larger governance structures, as highlighted by agency theory Jensen and Meckling [24] and resource-based theory Barney [28]. Instead of strengthening the relationship between ESG performance and dividend policy, a larger board size appears to weaken it. Consequently, the hypothesis that board size would positively moderate this relationship is not supported by the empirical results; therefore, we reject hypotheses H1 and H1a.

Model (2) examines the effect of the interaction between CEO duality and ESG performance on dividend distribution. The results indicate that CEO duality is associated with a significant decrease in dividend payouts (DIVPS), suggesting that the concentration of decision-making power in a single individual may lead to a less favorable dividend policy for shareholders. Furthermore, the coefficient of the (ESG\*DUAL) interaction is negative and significant, indicating that although ESG practices generally have a positive impact on dividends, this effect is weakened in the presence of CEO duality. This confirms that concentrated decision-making power can limit the expected benefits of ESG initiatives. Hence, we reject hypotheses H2 and H2a. The result aligns with agency theory, Jensen and Meckling [24], which posits that such concentration may reduce the incentive to maximize shareholder interests, as well as stakeholder theory [2], which emphasizes the importance of balanced governance in addressing stakeholder concerns. These results match up pretty closely with what Ahmad et al. [17], Afzalur et al. [126], Beji et al. [55], and Lin and Nguyen [56] found in their studies.

The model (3) estimates the interaction between the presence of independent directors and ESG performance (ESG\*IND). This interactional relationship has a positive and significant effect on dividend distribution. This suggests that companies with strong ESG scores benefit more from the influence of independent directors, which strengthens the positive effect of CSR practices on dividend policy. According to agency theory, these directors improve oversight and transparency, while stakeholder theory highlights their role in balancing interests. Signaling theory also indicates that this ESG engagement sends a positive message to the markets. In summary, these results confirm that board independence strengthens the relationship between ESG practices and dividend distribution, thereby validating hypotheses H3 and H3a. The results are consistent with the research of Al Fahli [127]; Celentano, et al. [128]; Di Guida, et al. [129], and Agarwala, et al. [85].

Results of Model (4) show that the coefficient of the interaction variable (ESG\*BGD) is positive. This indicates that gender diversity slightly strengthens the positive effect of ESG practices on dividend distribution. This suggests that the presence of women on the board of directors enhances the impact of ESG practices. Having gender diversity on a team can boost the company's dividends, and this positive effect becomes even stronger when the organization also performs well in environmental, social, and governance (ESG) performance.

According to diversity theory and stakeholder theory, mixed governance improves decision-making and risk management. In summary, gender diversity, by enhancing the impact of ESG practices, confirms hypotheses H4 and H4a regarding the positive moderating effect on the relationship between CSR and dividend distribution. These results are consistent with the works of Ye et al. [91] and Wong et al. [92].

The findings of Model (5) relate to the interaction between ESG performance and board cultural diversity (ESG\*BCD). Results indicate that this interaction is not statistically significant. While cultural diversity has a significant positive impact on dividend distribution, this interaction does not exert a significant effect. According to agency theory, cultural differences complicate coordination and increase costs, which may limit the impact of ESG practices. Legitimacy theory suggests that the effectiveness of ESG practices depends on the context and stakeholders' perceptions. The results show that companies with greater cultural diversity generally distribute more dividends, but the interaction with ESG does not have a significant moderating effect. In conclusion, although both ESG performance and cultural diversity are individually beneficial, their combination does not significantly impact dividend distribution policy. Therefore, hypotheses H5 and H5a regarding a positive moderating effect of cultural diversity are rejected.

Model (6) analyzes the interaction between ESG performance and the financial expertise of board members (BSS) on dividend distribution. The results show that this interaction is significant. This implies that financial expertise slightly strengthens the positive effect of ESG practices on dividends. According to agency theory, financial expertise helps improve resource management, reduce information asymmetry, and promote more efficient profit allocation. Signaling theory explains that such a profile sends a positive signal to investors, while stakeholder theory emphasizes a balance between financial performance and social responsibility. Hence, hypotheses H6 and

H6a are accepted. Although financial expertise alone does not directly influence dividend policy, the interaction between ESG and financial expertise enhances the positive effect of ESG on distribution. This suggests that financial expertise enables a better understanding of ESG issues, leading to more favorable distribution decisions for shareholders. The results confirm the findings of prior studies by Naheed et al. [103], Lin and Nguyen [56], and Zhe et al. [104].

### 5.3. Robustness Check: The Use of an Index of Board Characteristics (BC\_Index)

To check the robustness of our results, we conducted a robustness test following the methods proposed by Ahamed and Mallick [130] and Sarma [131]. The characteristics of the board of directors were measured using an index. The construction of this index was explained in detail in subsection 3.2.2. The results are presented in Table 7. This table shows that the p-values for the Sargan test and the serial correlation test (AR2) are above 5%. Therefore, we did not reject the null hypothesis of correct model specification.

Table 7. Result of the estimation using the index of board characteristics (BC\_Index).

DIVPS	Coef.	Std. err.	Z	P>z
DIVPS (-1)	-0.009	0.001	-6.790	0.000***
ESG	0.014	0.004	3.500	0.000***
BC_ index	-12.298	0.441	-27.870	0.000***
ESG * BC_ index	0.436	0.009	50.470	0.000***
SIZE	0.039	0.008	5.020	0.000***
ROA	0.015	0.005	3.130	0.002***
Cons.	-1.683	0.205	-8.210	0.000***
AR(1)		-1.234		
Prob.		0.2172		
AR(2)		99559		
Prob		0.3195		
Sargan test		25.8339		
Prob.		0.9745		
Obs.		13750		

Note:

\*\*\* indicates a significance level of 1%.

ESG\*BC\_index indicates an interactional relationship between CSR and board characteristics.

The results of this model show that all the included variables have a statistically significant impact on the dependent variable, with p-values below 1%. The DIVPS (-1) coefficient indicates a significant negative relationship, suggesting that an increase of one unit in DIVPS from the previous year is associated with a decrease of 0.009 in the dependent variable. ESG also shows a positive and significant relationship, with an increase of one unit in the ESG indicator linked to an increase of 0.014 in the dependent variable. In contrast, the BC\_index has a negative coefficient (-12.298), indicating that an increase in the BC\_index leads to a decrease of 12.298 in dividend payouts. The interaction between ESG and the BC\_index is also significant, highlighting that the impact of ESG on the dependent variable is influenced by the BC\_index. The firm size, with a coefficient of 0.039, suggests a positive relationship. When a company's size increases by 1%, its dividend payouts tend to rise slightly, by approximately 0.039. Similarly, the company's profitability also shows a positive relationship, with a coefficient of 0.015, suggesting that more profitable firms are likely to distribute higher dividends. Overall, these findings support the reliability of our model and highlight the importance of the variables examined, demonstrating both direct effects and meaningful interactions.

### 6. CONCLUSION AND POLICY RECOMMENDATIONS

This paper aims to understand how board characteristics influence the relationship between CSR performance and dividend payments in European firms. To achieve this, we examined data from 1,376 companies across 23 European countries over the period 2014 to 2023, using the SGMM approach as our primary empirical method.

Empirical results suggest that board characteristics play a significant role in shaping a company's commitment to ESG issues and its dividend policies.

In summary, our study confirms that specific features of a company's board have a substantial impact on how ESG performance relates to dividend decisions, highlighting the importance of these characteristics in corporate behavior. Our findings validate the expected influence of certain board characteristics on the relationship between CSR performance and dividend distribution. The study demonstrates that a stronger commitment to CSR significantly affects dividend distribution. This engagement in CSR appears to enhance investor confidence and positively influence their evaluation of the company.

This paper addresses the gap in the relationship between CSR performance and dividend distribution. Additionally, it examines the significance of the moderating effect of board characteristics on this relationship. The findings emphasize its critical role in decision-making, as well as its effectiveness in providing valuable information to investors and stakeholders. Consequently, the interaction between CSR practices and board characteristics serves as a key mechanism for informing stakeholders about corporate social responsibility. Corporate governance refers to a company's decision-making body responsible for setting strategic priorities and objectives in various areas, including financial results that may influence dividend distribution. The board of directors has become an entity increasingly monitored by investors. Therefore, from a conceptual perspective, this research invites us to consider the implications of different board characteristics in the interaction between CSR and dividend distribution.

This analysis highlights the managerial implications of the empirical findings, offering stakeholders tools to evaluate future growth opportunities in a context where corporate governance and board characteristics are crucial. The research emphasizes the importance of the interaction between these characteristics and sustainable practices, providing insights for CEOs and governance bodies to reconsider board composition and promote ethical behavior, particularly through CSR. The findings presented in this paper carry substantial policy implications. They offer critical insights for policymakers and stakeholders. First, the results suggest that policymakers should promote stricter regulations regarding board transparency and diversity to enhance accountability and the integration of sustainable practices into corporate strategies. Second, they highlight the need for incentive-based policies to encourage CSR, particularly through tax benefits or certifications that recognize ethical and environmental commitments. Third, they call for a revision of regulatory frameworks to explicitly incorporate ESG criteria into corporate evaluation mechanisms, enabling a more effective allocation of public and private resources.

Although the results of this paper are interesting, this study has some limitations. First, it does not take into account national differences in CSR practices and corporate governance, and it is limited to a European sample due to data availability. Second, additional variables, such as macroeconomic and institutional factors, could be included in the econometric model to explore the relationship between CSR, dividend payouts, and the moderating role of board characteristics. Future research could address these limitations by expanding the sample to other regions and exploring additional factors influencing the relationship between CSR and dividend distribution, using data from third-party platforms for a more comprehensive analysis. Additionally, including other economic and institutional variables could improve the results of this study.

**Funding:** This research was supported by the Deanship of Scientific Research at Imam Mohammad Ibn Saud Islamic University (IMSIU) (Grant number: IMSIU-DDRSP2504).

Institutional Review Board Statement: Not applicable.

**Transparency:** The authors state that the manuscript is honest, truthful, and transparent, that no key aspects of the investigation have been omitted, and that any differences from the study as planned have been clarified. This study followed all writing ethics.

**Competing Interests:** The authors declare that they have no competing interests.

**Authors' Contributions:** All authors contributed equally to the conception and design of the study. All authors have read and agreed to the published version of the manuscript.

#### **REFERENCES**

- [1] A. B. Carroll, "Corporate social responsibility: Evolution of a definitional construct," *Business & Society*, vol. 38, no. 3, pp. 268-295, 1999. https://doi.org/10.1177/000765039903800303
- [2] R. E. Freeman, Strategic management: A stakeholder approach. Boston: Pitman Publishing, 1984.
- [3] Y. M. Fourati and M. Dammak, "Corporate social responsibility and financial performance: International evidence of the mediating role of reputation," *Corporate Social Responsibility and Environmental Management*, vol. 28, no. 6, pp. 1749-1759, 2021.
- [4] H. N. Quang and T. T. Le, "Role of corporate social responsibility on firm performance in an emerging economy: The mediating role of access to finance and business model innovation," *Cogent Business & Management*, vol. 10, no. 2, p. 2232585, 2023. https://doi.org/10.1080/23311975.2023.2232585
- [5] M. E. Porter and M. R. Kramer, "The link between competitive advantage and corporate social responsibility," Harvard Business Review, vol. 84, no. 12, pp. 78-92, 2006.
- [6] J. Elkington, Cannibals with forks: The triple bottom line of 21st century business. Oxford: Capstone, 1997.
- [7] I. Ioannou and G. Serafeim, "The consequences of mandatory corporate sustainability reporting," Harvard Business School Research Working Paper No. 11-100. Harvard Business School, 2019.https://doi.org/10.1093/oxfordhb/9780198802280.013.20
- [8] S. Boubaker, A. Cellier, R. Manita, and A. Saeed, "Does corporate social responsibility reduce financial distress risk?," *Economic Modelling*, vol. 91, pp. 835-851, 2020. https://doi.org/10.1016/j.econmod.2020.05.012
- [9] D. Ogachi and Z. Zoltan, "Corporate social responsibility and firm value protection," *International Journal of Financial Studies*, vol. 8, no. 4, p. 72, 2020. https://doi.org/10.3390/ijfs8040072
- [10] M. C. Jensen and W. H. Meckling, "Theory of the firm: Managerial behavior, agency costs and ownership structure. In Corporate governance." London: Gower, 2019, pp. 77-132.
- [11] M. Rodriguez-Fernandez, "Social responsibility and financial performance: The role of good corporate governance," BRQ Business Research Quarterly, vol. 19, no. 2, pp. 137-151, 2016. https://doi.org/10.1016/j.brq.2015.08.001
- [12] I. Shaikh, "Environmental, social, and governance (ESG) practice and firm performance: An international evidence," Journal of Business Economics and Management, vol. 23, no. 1, pp. 218–237, 2022.
- [13] F. Hayat, N. Hamza, and A. Tariq, "Corporate social responsibility and firm performance in the Indian context,"

  Journal of World Economy Transformations & Transitions, vol. 2, no. 4, pp. 1-19, 2022. https://doi.org/10.52459/jowett24160922
- [14] G. Gavana, P. Gottardo, and A. M. Moisello, "Board diversity and corporate social performance in family firms. The moderating effect of the institutional and business environment," *Corporate Social Responsibility and Environmental Management*, vol. 30, no. 5, pp. 2194–2218, 2023. https://doi.org/10.1002/csr.2478
- [15] S. Finkelstein and R. A. D'aveni, "CEO duality as a double-edged sword: How boards of directors balance entrenchment avoidance and unity of command," *Academy of Management Journal*, vol. 37, no. 5, pp. 1079-1108, 1994.
- [16] A. Cheung, M. Hu, and J. Schwiebert, "Corporate social responsibility and dividend policy," *Accounting & Finance*, vol. 58, no. 3, pp. 787-816, 2018. https://doi.org/10.1111/acfi.12238
- [17] N. B. J. Ahmad, A. Rashid, and J. Gow, "CEO duality and corporate social responsibility reporting: Evidence from Malaysia," *Corporate Ownership and Control*, vol. 14, no. 2, pp. 69-81, 2017. https://doi.org/10.22495/cocv14i2art7
- [18] C. Post and K. Byron, "Women on boards and firm financial performance: A meta-analysis," *Academy of management Journal*, vol. 58, no. 5, pp. 1546-1571, 2015. https://doi.org/10.5465/amj.2013.0319
- [19] A. Khan, M. K. Yilmaz, and M. Aksoy, "Does board demographic diversity affect the dividend payout policy in Turkey?," *EuroMed Journal of Business*, vol. 19, no. 2, pp. 276-297, 2024. https://doi.org/10.1108/EMJB-01-2022-0019
- [20] D. Yermack, "Higher market valuation of companies with a small board of directors," *Journal of Financial Economics*, vol. 40, no. 2, pp. 185-211, 1996. https://doi.org/10.1016/0304-405X(95)00844-5

- [21] E. Diamastuti, M. Muafi, A. Fitri, and N. E. Faizaty, "The role of corporate governance in the corporate social and environmental responsibility disclosure," *The Journal of Asian Finance, Economics and Business*, vol. 8, no. 1, pp. 187-198, 2021. https://doi.org/10.13106/jafeb.2021.vol8.no1.187
- [22] F. Lakhal, C. Kuzey, A. Uyar, and A. S. Karaman, "The relationship between dividend payout and corporate social responsibility: The moderating effect of shareholder friendliness and board monitoring," *Journal of Cleaner Production*, vol. 394, p. 136297, 2023. https://doi.org/10.1016/j.jclepro.2023.136297
- [23] M. Farooq, A. Ahmed, I. Khan, and M. Munir, "Impact of dividend policy on corporate social responsibility: The role of board governance mechanism," *Journal of Global Responsibility*, vol. 16, no. 1, pp. 52-77, 2025. https://doi.org/10.1108/JGR-08-2023-0135
- [24] M. C. Jensen and W. H. Meckling, "Theory of the firm: Managerial behavior, agency costs, and ownership structure,"

  Journal of Financial Economics, vol. 3, no. 4, pp. 305–360, 1976. https://doi.org/10.1016/0304-405X(76)90026-X
- [25] J. Harford, S. A. Mansi, and W. F. Maxwell, "Corporate governance and firm cash holdings in the US," *Journal of Financial Economics*, vol. 87, no. 3, pp. 535-555, 2008. https://doi.org/10.1016/j.jfineco.2007.04.002
- [26] S. Bhattacharya, "Imperfect information, dividend policy, and" the bird in the hand" fallacy," *The Bell Journal of Economics*, vol. 10, no. 1, pp. 259-270, 1979. https://doi.org/10.2307/3003330
- [27] M. Orlitzky, F. L. Schmidt, and S. L. Rynes, "Corporate social and financial performance: A meta-analysis," Organization Studies, vol. 24, no. 3, pp. 403-441, 2003. https://doi.org/10.1177/0170840603024003910
- [28] J. Barney, "Firm resources and sustained competitive advantage," *Journal of Management*, vol. 17, no. 1, pp. 99-120, 1991. https://doi.org/10.1177/014920639101700108
- [29] A. B. Carroll, "The pyramid of corporate social responsibility: Toward the moral management of organizational stakeholders," *Business Horizons*, vol. 34, no. 4, pp. 39-48, 1991. https://doi.org/10.1016/0007-6813(91)90005-G
- [30] M. E. Porter and M. R. Kramer, "Strategy & society: The link between competitive advantage and corporate social responsibility," *Harvard Business Review*, vol. 84, no. 12, pp. 78–92, 2006.
- [31] J. Lintner, "Distribution of incomes of corporations among dividends, retained earnings, and taxes," *The American economic review*, vol. 46, no. 2, pp. 97-113, 1956.
- [32] M. H. Miller and F. Modigliani, "Dividend policy, growth, and the valuation of shares," *The Journal of Business*, vol. 34, no. 4, pp. 411-433, 1961. https://doi.org/10.1086/2944442
- [33] N. Dahlia, "Corporate social responsibility and dividend policy: Evidence from emerging markets," *Journal of Finance and Sustainability*, vol. 10, no. 2, pp. 145–160, 2023.
- [34] M. F. Sheikh, O. Masood, and Y. B. Tariq, "Corporate social responsibility and dividend payout: Evidence from Pakistan," Cogent Business & Management, vol. 8, no. 1, p. 1930898, 2021.
- [35] N. J. Dewasiri and N. Abeysekera, "Corporate social responsibility and dividend policy in Sri Lankan firms: A data triangulation approach," *Journal of Public Affairs*, vol. 22, no. 1, p. e2283, 2022. https://doi.org/10.1002/pa.2283
- [36] E. Abdel-Wanis, "Corporate social responsibility, corporate life cycle, and dividend policy," *Journal of Accounting, Business and Management*, vol. 27, no. 2, pp. 101-115, 2020. https://doi.org/10.31966/jabminternational.v27i2.703
- [37] M. Samet and A. Jarboui, "How does corporate social responsibility contribute to investment efficiency?," *Journal of Multinational Financial Management*, vol. 40, pp. 33-46, 2017.
- [38] V. Ongsakul, P. Jiraporn, and S. Treepongkaruna, "Does managerial ownership influence corporate social responsibility (CSR)? The role of economic policy uncertainty," *Accounting & Finance*, vol. 61, no. 1, pp. 763-779, 2021. https://doi.org/10.1111/acfi.12592
- [39] S. Chouaibi, M. Rossi, D. Siggia, and J. Chouaibi, "Exploring the moderating role of social and ethical practices in the relationship between environmental disclosure and financial performance: Evidence from ESG companies,"

  Sustainability, vol. 14, no. 1, p. 209, 2021. https://doi.org/10.3390/su14010209
- [40] A. Saeed and F. Zamir, "How does CSR disclosure affect dividend payments in emerging markets?," *Emerging Markets Review*, vol. 46, p. 100747, 2021. https://doi.org/10.1016/j.ememar.2020.100747

- [41] R. La Porta, F. Lopez-de-Silanes, A. Shleifer, and R. Vishny, "Investor protection and corporate governance," *Journal of financial economics*, vol. 58, no. 1-2, pp. 3-27, 2000. https://doi.org/10.1016/S0304-405X(00)00065-9
- [42] M. Azeem and R. Kouser, "Impact of quality corporate governance on firm performance: A ten year perspective," Pakistan Journal of Commerce and Social Sciences, vol. 7, no. 3, pp. 656-670, 2013.
- [43] C. T. Jie and N. A. M. Hasan, "Determinants of corporate social responsibility (CSR) and intrinsic job motivation," Malaysian Journal of Social Sciences and Humanities, vol. 1, no. 2, pp. 25-35, 2016.
- [44] O. R. Uwuigbe, "Corporate governance and financial performance of banks: A study of listed banks in Nigeria," Covenant Journal of Business and Social Sciences, vol. 2, no. 1, pp. 1–13, 2011.
- [45] D. Cormier, S. Demaria, and M. Magnan, "Beyond earnings: Do EBITDA reporting and governance matter for market participants?," *Managerial Finance*, vol. 43, no. 2, pp. 193-211, 2017. https://doi.org/10.1108/MF-07-2016-0205
- [46] S. A. R. Rizvi, "Board characteristics and firm performance: Evidence from emerging markets," *Corporate Governance: The International Journal of Business in Society*, vol. 23, no. 1, pp. 123–138, 2023.
- [47] M. Belkhir, "Board of directors' size and performance in the banking industry," *International Journal of Managerial Finance*, vol. 5, no. 2, pp. 201-221, 2009. https://doi.org/10.1108/17439130910947903
- [48] T. T. Y. Alabdullah, E. R. Ahmed, and M. Muneerali, "Effect of board size and duality on corporate social responsibility: What has improved in corporate governance in Asia?," *Journal of Accounting Science*, vol. 3, no. 2, pp. 121-135, 2019. https://doi.org/10.21070/jas.v3i2.2810
- [49] M. Al-Saidi, "Boards of directors and firm performance: A study of non-financial listed firms on the Kuwait Stock Exchange," *Corporate Ownership & Control*, vol. 18, no. 2, pp. 40-47, 2021. https://doi.org/10.22495/cocv18i2art3
- [50] W. Jilani and J. Chouaibi, "To what extent does CEO behavior enhance risk-taking? A banking sector related evidence," *Scientific Annals of Economics and Business*, vol. 68, no. 3, pp. 309-332, 2021.
- [51] A. B. Awad, B. A. Khalaf, and A. Afzal, "The power of board size and gender diversity on the value of companies listed on emerging markets," *Corporate Law & Governance Review*, vol. 5, no. 2, pp. 128–139, 2023.
- [52] K. J. Lin, K. Karim, R. Hu, and S. Dunn, "Fifty shades of CEO duality: CEO personal risk preference, duality and corporate risk-taking," *Journal of Applied Accounting Research*, vol. 24, no. 3, pp. 425-441, 2023. https://doi.org/10.1108/JAAR-02-2022-0034
- [53] I.-M. García-Sánchez, N. Hussain, and J. Martínez-Ferrero, "An empirical analysis of the complementarities and substitutions between effects of CEO ability and corporate governance on socially responsible performance," *Journal of Cleaner Production*, vol. 215, pp. 1288-1300, 2019. https://doi.org/10.1016/j.jclepro.2019.01.130
- [54] A. Kalsie and S. M. Shrivastav, "Analysis of board size and firm performance: Evidence from NSE companies using panel data approach," *Indian Journal of Corporate Governance*, vol. 9, no. 2, pp. 148-172, 2016. https://doi.org/10.1177/0974686216666456
- [55] R. Beji, O. Yousfi, N. Loukil, and A. Omri, "Board diversity and corporate social responsibility: Empirical evidence from France," *Journal of Business Ethics*, vol. 173, pp. 133-155, 2021. https://doi.org/10.1007/s10551-020-04522-4
- [56] C.-C. Lin and T. P. Nguyen, "Board attributes and corporate social responsibility performance: evidence from Vietnam," *Cogent Business & Management*, vol. 9, no. 1, p. 2087461, 2022. https://doi.org/10.1080/23311975.2022.2087461
- [57] M. La Rocca, F. Fasano, T. La Rocca, and N. Neha, "Women in CEO duality and firm performance in Europe," *Journal of Management and Governance*, vol. 28, no. 1, pp. 177-214, 2024.
- [58] A. A. Oussii and M. F. Klibi, "Corporate governance and corporate social responsibility disclosure: Evidence from an emerging market," *Corporate Governance: The International Journal of Business in Society*, vol. 23, no. 2, pp. 345–361, 2023.
- [59] C. Wijethilake and A. Ekanayake, "CEO duality and firm performance: The moderating roles of CEO informal power and board involvement," *Journal of Business Research*, vol. 113, pp. 302–310, 2020.
- [60] J. Tang, "CEO duality and firm performance: The moderating roles of other executives and blockholding outside directors," *European Management Journal*, vol. 35, no. 3, pp. 362-372, 2017. https://doi.org/10.1016/j.emj.2016.05.003

- [61] P. Debnath, P. Das, N. Laskar, S. B. Khan, S. Dhand, and K. Kaushal, "CEO duality and firm performance: An empirical study on listed companies from an emerging market," *Corporate Governance and Organizational Behavior Review*, vol. 5, no. 2, pp. 194–202, 2021. https://doi.org/10.22495/cgobrv5i2sip7
- [62] R. Mubeen, D. Han, J. Abbas, and I. Hussain, "The effects of market competition, capital structure, and CEO duality on firm performance: a mediation analysis by incorporating the GMM model technique," *Sustainability*, vol. 12, no. 8, p. 3480, 2020. https://doi.org/10.3390/su12083480
- [63] F. Turrent, V. González, and R. García, "CEO duality and dividend policy in family firms: Evidence from Brazil and Chile," *Journal of Family Business Strategy*, vol. 11, no. 3, pp. 100–112, 2020.
- [64] A. El Ammari, "Do CEO duality and ownership concentration impact dividend policy in emerging markets? The moderating effect of crises period," *International Journal of Financial Studies*, vol. 9, no. 4, p. 62, 2021. https://doi.org/10.3390/ijfs9040062
- D. Sheng and H. A. Montgomery, "Chief executive officer duality and the relationship between firm performance and dividend payouts," *Journal of Governance and Regulation*, vol. 13, no. 1, pp. 96–110, 2024. https://doi.org/10.22495/jgrv13i1art9
- [66] A. Rashid, S. Shams, S. Bose, and H. Khan, "CEO power and corporate social responsibility (CSR) disclosure: does stakeholder influence matter?," *Managerial Auditing Journal*, vol. 35, no. 9, pp. 1279-1312, 2020.
- [67] O. Yousfi and R. Béji, CSR: What does board diversity bring to the table? In B. Orlando (Ed.), Corporate social responsibility. United Kingdom: IntechOpen, 2020.
- [68] M. C. Pucheta-Martínez and I. Gallego-Álvarez, "The role of CEO power on CSR reporting: The moderating effect of linking CEO compensation to shareholder return," *Sustainability*, vol. 13, no. 6, p. 3197, 2021. https://doi.org/10.3390/su13063197
- [69] A. El Arfaoui and N. Marchoud, "The impact of social responsibility on corporate financial performance: a systematic literature review," *International Journal of Accounting, Finance, Auditing, Management and Economics*, vol. 5, no. 10, pp. 110–128, 2024.
- [70] R. Mubeen, D. Han, J. Abbas, S. Álvarez-Otero, and M. S. Sial, "The relationship between CEO duality and business firms' performance: The moderating role of firm size and corporate social responsibility," *Frontiers in Psychology*, vol. 12, p. 669715, 2021. https://doi.org/10.3389/fpsyg.2021.669715
- [71] R. Bhaskar, S. Bansal, and D. K. Pandey, "CEO duality and corporate social responsibility: A moderation effect of founder CEO," Research in International Business and Finance, vol. 71, p. 102410, 2024. https://doi.org/10.1016/j.ribaf.2024.102410
- [72] A. Arora, "Do independent directors enhance better corporate governance in companies in India?," *Public Administration and Policy: An Asia-Pacific Journal*, vol. 27, no. 2, pp. 154-166, 2024. https://doi.org/10.1108/PAP-05-2023-0057
- [73] J. Chang, "The role of independent directors in ensuring good corporate governance," Frontiers in Business Economics and Management, vol. 12, no. 1, pp. 7–11, 2023. https://doi.org/10.54097/fbem.v12i1.13618
- [74] A. Arora and T. K. Soni, "An optimal proportion for independent directors in the boardroom: An empirical study," Business Perspectives and Research, p. 22785337231170491, 2023. https://doi.org/10.1177/22785337231170491
- [75] E. F. Fama and M. C. Jensen, "Separation of ownership and control," *The Journal of Law and Economics*, vol. 26, no. 2, pp. 301-325, 1983. https://doi.org/10.1086/467037
- [76] N. Narang, S. Gupta, and N. Tripathy, "Dynamics of corporate governance and dividend policy alliance: A metaanalytical approach," *Corporate Governance: The International Journal of Business in Society*, vol. 25, no. 2, pp. 180-210, 2025. https://doi.org/10.1108/CG-05-2023-0183
- [77] S. Riaz, Y. Liu, and M. I. Ahmad, "Dividend policy and corporate governance perspective," *Accounting and Finance Research*, vol. 5, no. 3, pp. 77-86, 2016.

- [78] B. Alshabibi, S. Pria, and K. Hussainey, "Does board structure drive dividends payout? Evidence from the Sultanate of Oman," *Corporate Ownership and Control*, vol. 18, no. 4, pp. 218-230, 2021. https://doi.org/10.22495/cocv18i4art15
- [79] E. Kilincarslan, "The influence of board independence on dividend policy in controlling agency problems in family firms," *International Journal of Accounting & Information Management*, vol. 29, no. 4, pp. 552-582, 2021. https://doi.org/10.1108/IJAIM-03-2021-0056
- [80] M. S. Shahid, F. Gul, and M. Rizwan, "Ownership structure, board size, board composition and dividend policy: New evidence from two emerging markets," *IBT Journal of Business Studies*, vol. 2, no. 2, pp. 25–36, 2016.
- [81] O. Al Farooque, A. Hamid, and L. Sun, "Does governance structure have a say on dividends? Evidence from Australian listed firms", SSRN Working Paper No. 3529160, 2020.
- [82] B. K. Al Shabibi and G. Ramesh, "An empirical study on the determinants of dividend policy in the UK," *International Research Journal of Finance and Economics*, vol. 80, no. 12, pp. 105-124, 2011.
- [83] U. O. Ranti, "Determinants of dividend policy: A study of selected listed firms in Nigeria," *Manager*, no. 17, pp. 107-119, 2013.
- [84] R. Islam, M. Ali, and E. French, "Director independence and its influence on corporate social responsibility decisions and performance," *Social Responsibility Journal*, vol. 19, no. 10, pp. 1917-1934, 2023. https://doi.org/10.1108/SRJ-08-2022-0333
- [85] N. Agarwala, R. Pareek, and T. N. Sahu, "Does board independence influence CSR performance? A GMM-based dynamic panel data approach," *Social Responsibility Journal*, vol. 19, no. 6, pp. 1003-1022, 2022. https://doi.org/10.1108/SRJ-10-2020-0433
- [86] A. Al Fadli, R. Basiruddin, and R. Rusmin, "Corporate governance mechanisms and corporate social responsibility disclosure in Gulf cooperation council (GCC) countries," *Social Responsibility Journal*, vol. 16, no. 4, pp. 495–517, 2020.
- [87] H. Li and P. Chen, "Board gender diversity and firm performance: The moderating role of firm size," *Business Ethics: A European Review*, vol. 27, no. 4, pp. 294–308, 2018. https://doi.org/10.1111/beer.12188
- [88] R. S. Vinjamury, Corporate board gender diversity and dividend decisions: Evidence from India. In M. K. Hassan, M. Rabhi, & C. Viverita (Eds.), Contemporary issues in financial economics: Evidence from emerging economies. United Kingdom: Routledge, 2023.
- Z. Anum, A. Nasir, M. B. Hossain, T. S. Jagirani, F. Imran, and M. Garai Fodor, "A nexus between corporate gender diversity and dividend payout: Evidence from Asia pacific," *Journal of Infrastructure, Policy and Development*, vol. 8, no. 6, p. 3172, 2024. https://doi.org/10.24294/jipd.v8i6.3172
- [90] V. Q. Trinh, N. D. Cao, L. H. Dinh, and H. N. Nguyen, "Boardroom gender diversity and dividend payout strategies: Effects of mergers deals," *International Journal of Finance & Economics*, vol. 26, no. 4, pp. 6014-6035, 2021. https://doi.org/10.1002/ijfe.2106
- [91] D. Ye, J. Deng, Y. Liu, S. H. Szewczyk, and X. Chen, "Does board gender diversity increase dividend payouts? Analysis of global evidence," *Journal of Corporate Finance*, vol. 58, pp. 1-26, 2019. https://doi.org/10.1016/j.jcorpfin.2019.04.002
- [92] W. C. Wong, H. Batool, and M. H. Shakil, "Gender diversity on corporate boards and firm performance: Evidence from Malaysia," *Corporate Governance: The International Journal of Business in Society*, vol. 21, no. 6, pp. 1019–1038, 2021.
- [93] M. Gharbi and A. Jarboui, "The moderating effect of dividend policy on the relationship between corporate social responsibility and financial performance: Evidence from french context," *International Journal of Economics and Business Administration*, vol. 11, no. 2, pp. 97-114, 2023. https://doi.org/10.35808/ijeba/811
- [94] E. García-Meca, F. J. López-Iturriaga, and D. J. Santana-Martín, "Board gender diversity and dividend payout: The critical mass and the family ties effect," *International Review of Financial Analysis*, vol. 79, p. 101973, 2022. https://doi.org/10.1016/j.irfa.2021.101973
- [95] O. Dodd, B. Frijns, and A. Garel, "Cultural diversity among directors and corporate social responsibility," *International Review of Financial Analysis*, vol. 83, p. 102337, 2022. https://doi.org/10.1016/j.irfa.2022.102337

- [96] K. Aliani, F. Hamza, N. Alessa, H. Borgi, and K. Albitar, "ESG disclosure in G7 countries: Do board cultural diversity and structure policy matter?," *Corporate Social Responsibility and Environmental Management*, vol. 31, no. 4, pp. 3031-3042, 2024. https://doi.org/10.1002/csr.2733
- [97] R. B. Adams and D. Ferreira, "Women in the boardroom and their impact on governance and performance," *Journal of Financial Economics*, vol. 94, no. 2, pp. 291-309, 2009. https://doi.org/10.1016/j.jfineco.2008.10.007
- [98] I. Laksmana, M. Harjoto, and R. Lee, "Board diversity and corporate social responsibility," *Journal of business ethics*, vol. 132, no. 4, pp. 641-660, 2015. https://doi.org/10.1007/s10551-014-2343-0
- [99] H. Khan and A. K. Yilmaz, "Board diversity and corporate governance: A review of literature and agenda for future research," *Corporate Governance: The International Journal of Business in Society*, vol. 22, no. 7, pp. 1447–1465, 2022.
- [100] M. K. Yilmaz, U. Hacioglu, E. Tatoglu, M. Aksoy, and S. Duran, "Measuring the impact of board gender and cultural diversity on corporate governance and social performance: Evidence from emerging markets," *Economic Research*, vol. 36, no. 2, p. 2106503 2023. https://doi.org/10.1080/1331677X.2022.2106503
- [101] M. A. Naseem, R. Ali, and R. Ur Rehman, "Is there a mediating role of corporate social responsibility between board independence, board diversity, and dividend payouts decision?," *Gender in Management: An International Journal*, vol. 39, no. 2, pp. 291-308, 2024. https://doi.org/10.1108/GM-10-2022-0327
- [102] A. Abdulqader, Y. Wan Sallha, K. Fazida, and K. Mohd, "Board diversity and dividend policy: A review of the literature," in *AIP Conference Proceedings*, 2024.
- [103] R. Naheed, A. AlHares, Y. Shahab, and R. Naheed, "Board's financial expertise and corporate social responsibility disclosure in China," *Corporate Governance: The International Journal of Business in Society*, vol. 21, no. 4, pp. 716-736, 2021. https://doi.org/10.1108/CG-08-2020-0329
- [104] L. Zhe, B. Wang, and D. Zhou, "Financial experts of top management teams and corporate social responsibility: Evidence from China," *Review of Quantitative Finance and Accounting*, vol. 59, no. 4, pp. 1335-1386, 2022. https://doi.org/10.1007/s11156-022-01077-5
- [105] E. K. Thompson and A. S. Manu, "The impact of board composition on the dividend policy of US firms," *Corporate Governance: The International Journal of Business in Society*, vol. 21, no. 5, pp. 737-753, 2021. https://doi.org/10.1108/CG-05-2020-0182
- [106] B. Sarwar, M. Xiao, M. Husnain, and R. Naheed, "Board financial expertise and dividend-paying behavior of firms: New insights from the emerging equity markets of China and Pakistan," *Management Decision*, vol. 56, no. 9, pp. 1839-1868, 2018. https://doi.org/10.1108/MD-11-2017-1111
- [107] W. Wu, J. Hou, and S. Cheng, "Board financial expertise and dividend policy: Evidence from an emerging market,"

  International Review of Economics & Finance, vol. 73, pp. 220–238, 2021.
- [108] S. El Ghoul, O. Guedhami, and C. C. Y. Kwok, "Corporate social responsibility and asset prices: The role of board financial expertise," *Journal of Corporate Finance*, vol. 42, pp. 1-19, 2017.
- [109] L. J. Gitman and C. J. Zutter, Principles of managerial finance, 14th ed. Boston, MA: Pearson, 2015.
- [110] R. Sullivan, Valuing corporate responsibility: How do investors really use corporate responsibility information?, 1s ed. UK: Routledge, 2011.
- [111] K. Rao and C. Tilt, "Board diversity and CSR reporting: An Australian study," *Meditari Accountancy Research*, vol. 24, no. 2, pp. 182-210, 2016. https://doi.org/10.1108/MEDAR-08-2015-0052
- [112] M. A. Zaid, M. Wang, and S. T. Abuhijleh, "The effect of corporate governance practices on corporate social responsibility disclosure: Evidence from Palestine," *Journal of Global Responsibility*, vol. 10, no. 2, pp. 134-160, 2019. https://doi.org/10.1108/JGR-10-2018-0053
- [113] M. Adib and X. Zhang, "The effect of CEO duality on corporate social performance: Evidence from South Africa," African Journal of Business Management, vol. 13, no. 10, pp. 236–251, 2019.
- [114] A. Al-Mamun, "Board diversity and corporate social responsibility reputation: Evidence from Bangladesh," *Corporate Governance: The International Journal of Business in Society*, vol. 20, no. 4, pp. 701–717, 2020.

- [115] W. Ben-Amar, M. Chang, and P. McIlkenny, "Board gender diversity and corporate response to sustainability initiatives: Evidence from the carbon disclosure project," *Journal of Business Ethics*, vol. 142, no. 2, pp. 369-383, 2017. https://doi.org/10.1007/s10551-015-2759-1
- [116] N. Cucari, S. Esposito De Falco, and B. Orlando, "Diversity of board of directors and environmental social governance: Evidence from Italian listed companies," *Corporate Social Responsibility and Environmental Management*, vol. 25, no. 3, pp. 250–266, 2018. https://doi.org/10.1002/csr.1452
- [117] R. Krause, M. Semadeni, and A. A. Cannella Jr, "CEO duality: A review and research agenda," *Journal of Management*, vol. 40, no. 1, pp. 256-286, 2014.
- [118] E. García-Meca, I.-M. García-Sánchez, and J. Martínez-Ferrero, "Board diversity and its effects on bank performance:

  An international analysis," *Journal of Banking & Finance*, vol. 53, pp. 202-214, 2015. https://doi.org/10.1016/j.jbankfin.2014.12.002
- [119] N. Vafaei, R. A. Ribeiro, and L. M. Camarinha-Matos, "Normalization techniques for multi-criteria decision making: Analytical hierarchy process case study. In Technological innovation for cyber-physical systems," in 7th IFIP WG 5.5/SOCOLNET Advanced Doctoral Conference on Computing, Electrical and Industrial Systems, DoCEIS 2016, Costa de Caparica, Portugal, April 11–13, 2016, Proceedings 7, Springer International Publishing., 2016, pp. 261–269.
- [120] Y. Chang, K.-T. Wu, S.-H. Lin, and C.-J. Lin, "Board gender diversity and corporate social responsibility," *International Journal of Corporate Social Responsibility*, vol. 9, no. 1, p. 7, 2024. https://doi.org/10.1186/s40991-024-00095-x
- [121] N. Bidi, "Board gender diversity and firm performance: Evidence from French firms," Corporate Governance: The International Journal of Business in Society, vol. 20, no. 2, pp. 403–424, 2020.
- [122] R. Blundell and S. Bond, "Initial conditions and moment restrictions in dynamic panel data models," *Journal of Econometrics*, vol. 87, no. 1, pp. 115–143, 1998.
- [123] M. J. Flannery and K. W. Hankins, "Estimating dynamic panel models in corporate finance," *Journal of Corporate Finance*, vol. 19, pp. 1–19, 2013.
- [124] M. B. Wintoki, J. S. Linck, and J. M. Netter, "Endogeneity and the dynamics of internal corporate governance," *Journal of Financial Economics*, vol. 105, no. 3, pp. 581–606, 2012. https://doi.org/10.1016/j.jfineco.2012.03.005
- [125] N. G. Damodar and D. C. Porter, "Basic econometrics." Singapore: McGraw-Hill, 2004, p. 858.
- [126] R. M. Afzalur, A. Mahbub, and M. H. Kabir, "Corporate governance and dividend policy: A panel data analysis of Australian companies," *Journal of Risk and Financial Management*, vol. 13, no. 10, pp. 1-18, 2020.
- [127] A. M. A. Al Fahli, "The impact of board characteristics on corporate social responsibility disclosure: Evidence from Oman," *International Journal of Innovation, Creativity and Change*, vol. 15, no. 8, pp. 1112–1133, 2021.
- [128] A. Celentano, L. Lepore, S. Pisano, G. D'Amore, and F. Alvino, "The joint effect of board independence and CSR committee on CSR disclosure: Evidence from Italian listed companies," *Corporate Ownership and Control*, vol. 17, no. 4, pp. 319–328, 2020. https://doi.org/10.22495/cocv17i4siart9
- [129] C. Di Guida, A. Celentano, F. Mirone, and G. D'Amore, "The role of board composition in CSR disclosure: An empirical analysis in four European countries," *Corporate Governance and Organizational Behavior Review*, vol. 6, no. 3, pp. 74–86, 2022. https://doi.org/10.22495/cgobrv6i3p7
- [130] M. M. Ahamed and S. K. Mallick, "Is financial inclusion good for bank stability? International evidence," *Journal of Economic Behavior & Organization*, vol. 157, pp. 403–427, 2019. https://doi.org/10.1016/j.jebo.2017.07.027
- [131] M. Sarma, "Index of financial inclusion," Working Paper No. 215, New Delhi: Indian Council for Research on International Economic Relations, 2008.

Views and opinions expressed in this article are the views and opinions of the author(s), Journal of Asian Scientific Research shall not be responsible or answerable for any loss, damage or liability etc. caused in relation to/arising out of the use of the content.